Author's response to reviews

Title: Targeted versus tailored multimedia patient engagement to enhance depression recognition and treatment in primary care: randomized controlled trial protocol

Authors:

Daniel J Tancredi (djtancredi@ucdavis.edu)
Christina K Slee (cakuenneth@ucdavis.edu)
Anthony Jerant (anthony.jerant@ucdmc.ucdavis.edu)
Peter Franks (pfranks@ucdavis.edu)
Jasmine Nettiksimmons (janettik@ucdavis.edu)
Camille Cipri (cscipri@ucdavis.edu)
Dustin Gottfeld (dustin.gottfeld@ucdmc.ucdavis.edu)
Julia Huerta (jhuerta1@jhu.edu)
Mitchell D Feldman (mitchell.feldman@ucsf.edu)
Maja Jackson-Triche (maja.jackson-Triche@va.gov)
Steven Kelly-Reif (steve.kelly-reif@kp.org)
Andrew Hudnut (HudnutA@sutterhealth.org)
Sarah Olson (Sarah.Olson@ucsf.edu)
Janie Shelton (janie.shelton@gmail.com)
Richard L Kravitz (rlkravitz@ucdavis.edu)

Version: 4 Date: 20 February 2013

Author's response to reviews: see over